Author's response to reviews

Title: Blood pressure control and antihypertensive pharmacotherapy patterns in a hypertensive population of Eastern Central Region of Portugal

Authors:

Manuel P Morgado (manuelaugustomorgado@gmail.com)
Sandra A Rolo (sandracristinarolo@clix.pt)
Luísa Pereira (lpereira@mat.ubi.pt)
Miguel Castelo-Branco (mcbranco@fcsaude.ubi.pt)

Version: 4 Date: 23 December 2010

Author's response to reviews: see over
Dear Editor,

We are submitting our manuscript entitled "Blood pressure control and antihypertensive pharmacotherapy patterns in a hypertensive population of Eastern Central Region of Portugal", with the requested formatting changes duly addressed. We hope that the manuscript is now in agreement to the journal (BMC Health Services Research) style.

Please, see bellow a point-by-point description of the changes made to the most recent version of our manuscript (accessed at the link provided by you). Please let us know if further corrections are needed.

We look forward to hearing from you.

Kindest regards,

Manuel Morgado, PharmD, MSc
University of Beira Interior, Health Sciences Research Centre,
Av. Infante D. Henrique, 6200-506 Covilhã
Portugal

Tel.: +351 275329072, Fax: +351 275329099
E-mail: manuelaugustomorgado@gmail.com
www.fcsaude.ubi.pt
Point-by-point description of the changes made to the most recent version of our manuscript (accessed at the link provided by you)

1- Table 3 was included in the manuscript in a tables section following the references, because we want the table to be visible within the final published manuscript.

2- All highlighting /tracking were removed from the manuscript.

3- Authors' contributions was formatted into a paragraph rather than a list or in bullet points.

4- All visible vertical lines from our tables and the diagonal line from table 2 were removed.